Author's response to reviews

Title: Eschar and neck lymphadenopathy caused by francisella tularensis after tick bite: a case report.

Authors:

Sophie Edouard (sophie.edouard@ap-hm.fr)
Khira Gonin (khira.gonin@ch-creil.fr)
Yves Turc (yves.turc@ch-creil.fr)
Emmanouil Angelakis (Emmanouil.angelakis@univmed.fr)
Cristina Socolovschi (cristina.socolovschi@univmed.fr)
Didier Raoult (didier.raoult@gmail.com)

Version: 2 Date: 7 June 2010

Author's response to reviews: see over
Dear Editor,

Please find the revised version of “Eschar and neck lymphadenopathy caused by Francisella tularensis after tick bite: a case report”. As required find below the formatting changes:

Line 31 and 55: We include the ethnicity and sex of the patient in the case presentation section of the manuscript and in the abstract. We replace “A 4- years old child” by “A 4-years old French boy”

Line 109: We confirm that we have obtained written informed consent from the parents of the patient for publication of the case because the patient is a minor. The consent section is modified accordingly.
Line 56: In order to protect the patient’s identity, we remove the dates of treatment. We suppressed “on May 2009”

Sincerely yours.

Didier Raoult, PhD, MD, corresponding authors.

Email: didier.raoult@gmail.com.